Author's response to reviews

Title: Prevalence of high frequency hearing loss consistent with noise exposure among people working with sound systems and general population in Brazil: a cross-sectional study

Authors:

Regina P El Dib (re.lucci@terra.com.br)
Edina MK Silva (edinaksilva@terra.com.br)
Jose F Morais (jfmorais@pucsp.br)
Virginia FM Trevisani (vmoca@uol.com.br)

Version: 9 Date: 27 April 2008

Author's response to reviews:

No cover letter for this version, only Formatting Changes for MS: 6959209031564007.